Author's response to reviews

Title: Follow-up after surgery with ventilation tube in the tympanic membrane: Implementing guidelines in primary care and hospital. A retrospective study

Authors:

Bjarne Austad (bjarne.austad@ntnu.no)
Irene Hetlevik (irene.hetlevik@ntnu.no)
Vegard Bugten (vegard.bugtne@stolav.no)
Siri Wennberg (siri.wennberg@stolav.no)
Anita H Olsen (anita.helene.olsen@stolav.no)
Anne-Sofie Helvik (anne-sofie.helvik@ntnu.no)

Version: 3 Date: 28 September 2012

Author's response to reviews:

In this version I have answered your two questions:

1. The consent from the parents/guardians (in the manuscript)
2. Enclosed the original questionnaire (in Norwegian)

Kind regards, Bjarne Austad